Author’s response to reviews

Title: Reablement in community-dwelling adults: study protocol for a randomised controlled trial

Authors:

Hanne Tuntland (htu@hib.no)
Birgitte Espehaug (Birgitte.Espehaug@hib.no)
Oddvar Forland (Oddvar.Forland@hib.no)
Astri D Hole (Astri.Drange.Hole@hib.no)
Egil Kjerstad (Egil.Kjerstad@uni.no)
Ingvild Kjeken (ingvild.kjeken@diakonsyk.no)

Version: 2 Date: 12 December 2014

Author’s response to reviews: see over
12 December 2014

Dear editor

MS: 47610965514724865 – Study protocol

Please find enclosed the revised version of the original article: “Reablement in community-dwelling adults: study protocol for a randomised controlled trial”.

The following revisions are performed prior to resubmitting:

1. The funding and ethics documents are removed for additional files
2. The abstract is inserted into the main text of the manuscript
3. The trial registration is changed to the official ClinicalTrials.gov Identifier
4. A flow diagram according to CONSORT guidelines is uploaded
5. I confirm that the protocol was peer reviewed prior to funding being granted
6. The manuscript contains required statements regarding ethics approval and consents from participants
7. The manuscript adheres to CONSORT guidelines for transparent reporting

However, the data sets will not be published (concerning availability of supporting data). Disseminating data sets was not planned from the beginning; hence ethics approval and participants’ consents are not obtained for this purpose.

Kind regards,

Hanne Tuntland
OT, Associate Professor